Author's response to reviews

Title: Transcript levels of Toll-like receptors 5, 8 and 9 correlate with inflammatory activity in Ulcerative Colitis

Authors:

Fausto Sanchez-Muñoz (fausto22@yahoo.com)
Gabriela C Fonseca-Camarillo (gabrielafaster@gmail.com)
Marco A Villeda-Ramírez (nmavilledar_snk@hotmail.com)
Elizabeth Miranda-Perez (mirandapme@yahoo.com.mx)
Edgar J Mendivil (ed-jair@hotmail.com)
Rafael Barreto-Zuñiga (barretozu@yahoo.com)
Misael Uribe (muribe@rtm.net.mx)
Rafael Bojail (bojraf@yahoo.com)
Aarón Domínguez-Lopez (aadominguezl@yahoo.com.mx)
Jesus K Yamamoto-Furusho (kazuofurusho@hotmail.com)

Version: 2 Date: 19 October 2010

Author's response to reviews: see over
REF: Submission of paper: Transcript levels of Toll-like receptors 5, 8 and 9 mRNA correlate with inflammatory activity in Ulcerative Colitis

Editor-in-Chief
Melissa Norton, MD
BMC Gastroenterology

Dear Dr.

Attached please find paper submitted for publication entitled Transcript levels of Toll-like receptors 5, 8 and 9 correlate with inflammatory activity in Ulcerative Colitis by Fausto Sánchez Muñoz, Gabriela Columba Fonseca Camarillo, Marco Antonio Villeda Ramírez, Elizabeth Miranda Perez, Edgar Jair Mendivil, Rafael Barreto Zuniga, Misael Uribe, Rafael Bojalil, Aarón Domínguez Lopez, Jesús K. Yamamoto Furusho.

Our manuscript had significant contribution of all authors, has not been submitted nor published in any other journal.

We submit this paper and considered suitable to BMC Gastroenterology because, we consider is a careless effort to assess all reported Toll like Receptors mRNA levels in a careful characterized sample of Ulcerative Colitis and Controls. Our effort also confirms the complex participation of Toll like receptor gene expression as part of Ulcerative Colitis, and shows that less studied TLR5, TLR8 and TLR9 in UC have important modulation in IBD and may be important as molecular markers.

Please inform us, as soon as possible if paper meets the established requirements because it acceptance is required to obtain PhD Degree. We also add that we agree with all Journal requirements

Yours sincerely

M. D. Ph. D. Jesus Kazuo Yamamoto Furusho
Chief of the Inflammatory Bowel Disease Clinic
Instituto Nacional de Ciencias Médicas y Nutrición Salvador Zubiran